

Clinical Commissioning Policy: Obesity surgery for children with severe complex obesity

Reference: NHS England: 16053/P



Directorate		
Medical	Operations and Information	Specialised Commissioning
Nursing	Trans. & Corp. Ops.	Commissioning Strategy
Finance		
Publications Gateway	Reference:	05527s
Document Purpose	Policy	
Document Name	Obesity surgery for children with severe complex obesity	
Author	Specialised Commissioning Team	
Publication Date	18 April 2017	
Target Audience	CCG Clinical Leaders, Care Trust CEs, Foundation Trust CEs, Medical Directors, Directors of PH, Directors of Nursing, NHS England Regional Directors, NHS England Directors of Commissioning Operations, Directors of Finance, NHS Trust CEs	
Additional Circulation List		
Description	Routinely Commissioned - NHS England will routinely commission this specialised treatment in accordance with the criteria described in this policy.	
Cross Reference	N/A	
Superseded Docs	Clinical Commissioning Policy: Complex and specialised obesity surgery	
(if applicable)	(all ages)	
Action Required	N/A	
Timing / Deadlines (if applicable)	N/A	
Contact Details for further information	england.specialisedcommissioning@nhs.net	

This is a controlled document. Whilst this document may be printed, the electronic version posted on the intranet is the controlled copy. Any printed copies of this document are not controlled. As a controlled document, this document should not be saved onto local or network drives but should always be accessed from the intranet.

Clinical Commissioning Policy: Obesity surgery for children with severe complex obesity

First published: April 2017

Prepared by NHS England Specialised Services Clinical Reference Group for Severe and Complex Obesity

Published by NHS England, in electronic format only.

Contents

Introduction	7
Definitions	8
Aims and Objectives	9
Epidemiology and Needs Assessment	10
Evidence base	10
Criteria for Commissioning	15
Patient Pathway	16
Governance Arrangements	18
Mechanism for Funding	19
Audit Requirements	19
Documents which have informed this Policy	19
Date of Review	20
ences	21
	Definitions Aims and Objectives Epidemiology and Needs Assessment Evidence base Criteria for Commissioning Patient Pathway Governance Arrangements Mechanism for Funding Audit Requirements Documents which have informed this Policy Date of Review

Policy Statement

NHS England will commission obesity surgery for children with severe complex obesity in accordance with the criteria outlined in this document. In creating this policy NHS England has reviewed this clinical condition and the options for its treatment. It has considered the place of this treatment in current clinical practice, whether scientific research has shown the treatment to be of benefit to patients, (including how any benefit is balanced against possible risks) and whether its use represents the best use of NHS resources. This policy document outlines the arrangements for funding of this treatment for the population in England.

Equality Statement

Promoting equality and addressing health inequalities are at the heart of NHS England's values. Throughout the development of the policies and processes cited in this document, we have:

- Given due regard to the need to eliminate discrimination, harassment and victimisation, to advance equality of opportunity, and to foster good relations between people who share a relevant protected characteristic (as cited under the Equality Act 2010) and those who do not share it; and
- Given regard to the need to reduce inequalities between patients in access to, and outcomes from healthcare services and to ensure services are provided in an integrated way where this might reduce health inequalities.

Plain Language Summary

About childhood and adolescent obesity

Obesity in children and adolescents is a major and growing health problem. Having a lot of body fat and being very overweight (called 'obesity') can lead to a number of serious problems such as:

- high blood pressure (called 'hypertension')
- a condition where the body produces insulin but does not use it effectively (called 'insulin resistance')
- a group of problems that can lead to heart disease, stroke and type 2 diabetes (called 'metabolic syndrome')
- lower life expectancy.

About current treatments

For most people, eating a healthy, reduced-calorie diet and exercising regularly is the most effective treatment for obesity. Some patients may benefit from psychological support from a trained healthcare professional, to help change the way they think about food and eating.

Specialist weight management programmes are also available, although they are often designed for adults. If lifestyle changes alone are not successful, treatment with medicines may be considered.

All of the above treatments are 'non-invasive' - they do not require an operation.

About the new treatment

If patients do not respond to the above 'non-invasive' therapies, obesity surgery may be considered. This refers to any surgical treatment or operation for obesity - this may include a gastric bypass, for example.

What we have decided

NHS England has carefully reviewed the evidence to treat childhood and adolescent obesity with obesity surgery. We have concluded that there is enough evidence to consider making the treatment available for a small number of highly selected children, 18 years and under with severe and complex obesity.

1 Introduction

This document describes the evidence that has been considered by NHS England in formulating clinical policy to routinely commission obesity surgery for children with severe and complex obesity for a small number of highly selected patients.

Obese children and adolescents are at an increased risk of developing various health problems, and are also more likely to become obese adults. Childhood obesity is associated with co-morbid conditions, commonly hypertension, obstructive sleep apnoea, insulin resistance, metabolic syndrome, non-alcoholic fatty liver disease and dyslipidaemia. The cost of obesity to society was estimated in 2007 to be £16 billion, and if rates continue to rise could reach up to £50 billion in 2050 (NICE CG189, 2014).

Obesity in children is currently managed predominately with lifestyle interventions, focusing on behavioural and dietary modifications, with evidence of short term success (Cochrane Review, 2009). Pharmacotherapy is less commonly used in adolescent patients: Cochrane review (2009) showed both orlistat and sibutramine in children greater than 12 years to be beneficial in reducing weight at 6 months. Some severely obese adolescents with significant and severe obesity-related comorbidities such as hypertension, fatty liver disease or uncontrolled diabetes, who have failed specialist multi-component, intensive, non-invasive weight management programmes, may benefit from a surgical approach.

Currently there is no evidence based care pathway in utilising obesity surgery in the paediatric/adolescent population. Primarily three types of obesity surgery are being commonly performed in the paediatric/adolescent population: laparoscopic Roux-en Y gastric bypass (RYGB); laparoscopic adjustable gastric banding (LAGB); and more recently sleeve gastrectomy (LSG).

2 Definitions

Obesity: In adults, obesity is commonly defined as a body mass index of 30 or more. For children, in the UK, the British 1990 growth reference charts are used to define weight status.

Body mass index (BMI): a measure (kg/m²) of whether someone is a healthy weight for their height.

BMI SD: standard deviation score indicates how many units (of the standard deviation) a child's BMI is above or below the average BMI value for their age group and sex. Also referred to as a z score.

Co-morbidity: the presence of one or more additional diseases co-occurring with a primary disease (synergistic or coincidental); or the effect of such additional disease (clinically dominant).

Dyslipidaemia or high cholesterol: means that there is an imbalance of fats (lipids), circulating in the blood stream. Cholesterol is a fatty substance the body uses to make hormones and metabolise food.

Obesity surgery: also known as bariatric surgery, any surgical treatment for obesity.

Laparoscopic adjustable gastric band (LAGB, or gastric banding): helps reduce the amount of food eaten. It acts like a belt around the top portion of the stomach, creating a small pouch. Patients feel full after eating only a small quantity of food. It is adjustable and reversible.

Roux–en–Y gastric bypass (RYGB): the most popular variation of gastric bypass operation conducted in the UK. During surgery, the top section of the stomach is divided off by a line of staples, creating a small 'pouch' stomach. A new exit from this pouch is made into a 'Y' loop from the small intestine so that food bypasses the old stomach and part (about 100-150cm) of the small intestine. The size of stomach

pouch and the length of small intestine that is bypassed are carefully calculated to ensure that patients will be able to eat enough for their body's needs at normal weight.

Laparoscopic sleeve gastrectomy (LSG): the sleeve gastrectomy reduces the size of the stomach by about 75%. It is divided vertically from top to bottom leaving a banana shaped stomach along the inside curve, and the pyloric valve at the bottom of the stomach, which regulates the emptying of the stomach into the small intestine, remains intact. This means that although smaller, the stomach function remains unaltered.

Gastric balloon: an intra-gastric balloon is a soft silicone balloon that is surgically implanted into the stomach. The balloon is filled with air or saline solution (sterile salt water), and so takes up some of the space in the stomach. This procedure is only temporary, and the balloon is usually removed after six months.

Models of care: a typical model for managing obesity is outlined as follows:

- Tier 4 Specialised Complex Obesity Services (including both medical management, obesity surgery and other elements of specialised multi-disciplinary team [MDT] care)
- Tier 3 An MDT to provide an intensive level of input to patients
- Tier 2 Primary Care with Community Interventions. The Tier 2 programme will typically be of 10-12 weeks duration. It can be provided by the NHS or commercially. Identification of co-morbidities can take place in either primary or secondary care. If necessary patients may be referred to Tier 3 for identification of co-morbidities. Patients with non-engagement or child protection issues can be transferred directly to Tier 4
- Tier 1 Primary Care and Community Advice

3 Aims and Objectives

This clinical policy aims to define NHS England's commissioning position on obesity surgery as part of the treatment pathway for children 18 years and under with severe and complex obesity.

The objective is to ensure evidence based commissioning with the aim of improving outcomes for children with severe and complex obesity.

4 Epidemiology and Needs Assessment

Childhood and adolescent obesity is a major and growing health problem and associated with comorbid conditions, commonly hypertension, obstructive sleep apnoea, insulin resistance, metabolic syndrome, non-alcoholic fatty liver disease and dyslipidaemia.

The prevalence of childhood obesity has been increasing, and in 2011, 3 in 10 children aged 2-15 years were found to be overweight or obese in the UK (NICE CG189). Rates of obesity surgery are also increasing: there was 1 operation in 2000 and 31 in 2009 (Jones Nielson et al., 2013). It is estimated that around 6-8 patients receive obesity surgery each year, based on an average across the 30-40 undertaken over the last 5 years and data from the National Bariatric Surgery Registry (2014) that shows that 23 primary operations for patients aged 12-17 were undertaken between 2011 - 2013.

5 Evidence base

NHS England has concluded that there is sufficient evidence to support a clinical policy for the routine commissioning of obesity surgery for children with severe and complex obesity for a small number of highly selected patients.

What is the clinical effectiveness of obesity surgery in children and adolescents? Is there any evidence for long term efficacy (more than 1 year? more than 5 years?)?

Overall, in the current literature there is evidence of clinical effectiveness for obesity surgery in adolescents (following skeletal maturity - Tanner Stage 4 and above),

predominately from non-Randomised Control Trial (RCT) studies (level 2 and 3 studies), with limited evidence about long term efficacy. There are limited studies on performing obesity surgery in younger children (level 3). There is insufficient evidence on selection criteria, indications, postoperative complications and long-term adverse effects of surgery. Although included in the literature search strategy, no evidence relating to duodenal switch procedures in adolescents could be found. The search strategy did not specifically include rare syndromes predisposing to adolescent obesity, but some pertinent information was found in the wider literature.

One RCT (level 1) by O'Brien et al. (2010) has been identified in the current literature, which evaluated LAGB with intensive lifestyle intervention (dietary and behavioural modification) in 50 obese adolescents aged 14-18. They found that LAGB resulted in substantial weight loss at two years, with a mean reduction of 34.6kg versus 3kg in the lifestyle group. They also observed improvements in health related quality of life. Twenty-eight percent of adolescents required revision surgery, removal or replacement of the band or replacement of the access port, a rate consistent with adult studies.

A recent meta-analysis (level -1) by Paulus et al. (2015) examined change in BMI one year post operatively, and reported on health related indices. The analysis included 23 studies (level 2 and 3), and found the mean BMI loss was -13.5kg/m², the greatest loss in the RYGB group (-17.2kg/m²) and smallest in the LAGB group (-10.5kg/m²). These findings were consistent with another meta-analysis that evaluated 37 studies (Black et al., 2013) (level -1), and found the mean BMI loss was greatest in the RYGB group (-16.6kg/m²), followed by LSG with 14.1kg/m² and LAGB with -11kg/m². Pedroso et al. (2015) (level -2) assessed LSG and LAGB in adolescent patients and at two year follow-up observed significantly greater percentage excess weight loss in the LSG group compared to the LAGB group (70.9% vs 35.5% respectively p=0.004). The Teen Longitudinal Assessment of Bariatric Surgery (Teen-LABS) study (Inge et al., 2015) showed that patients who underwent obesity surgery (RYGB and LSG) reported an overall decrease in mean weight of 27% and mean BMI decrease of 28% (BMI decreased from baseline from 53kg/m² to 38kg/m²) at 3 years post operatively. The mean weight loss of those patients who underwent RYGB was 28% compared to 26% in the LSG group at three

years. The study observed that at three years 26% of patients were no longer obese. At 3 years 2% of patients who underwent gastric bypass and 4% of those who underwent sleeve gastrectomy exceeded baseline weight.

Paulus et al. (2015) noted the overall poor quality of documentation of complications, with the majority of complications in the RYGB group involving nutrient deficiencies, hernia and wound infection. In LAGB the key complications were pouch dilatation, band slippage and port complications. Complications were rarely reported in LSG. Pedroso et al. (2015) observed that at 5 years the complication rate in the LAGB group was 23.4%, which included bowel obstruction, port leakage and band displacement. Follow-up at two years in the LSG group reported minimal overall complications. However, there was one death 12 days post LSG, as a result of mesenteric venous thrombosis. The Teen-LABS study (n=242) (level 3), which is an ongoing prospective study, evaluated outcomes within 30 days postoperatively (Inge et al., 2014). No mortality was recorded, 7.9% experienced major complications, 5% perioperative complications including one splenic injury, early reoperation for intestinal obstruction, bleeding or suspected gastrointestinal leak. 14.9% had minor complications including urinary tract infections, abdominal and gastrointestinal complaints including dehydration. The Teen-LABS study at 3 years post operatively (Inge et al. 2015) found 13% of patients had undergone one or more intra-abdominal procedure. Inge et al. also evaluated micronutrients, and found low ferritin levels were evident in 57% of patients (p<0.001), 16% of patients who undergone RYGB (p=0.008) and 8% of all patient being VitB12 deficient, at three years post-surgery. Case series of 345 patients Lennez et al. (2014) reported intraoperative complications rate of 0%-2.6%, and postoperative complications (18 months followup) rate of 9.1% to 2.5%. In this case series they found no difference in rates of complications at 18 months amongst the three surgical procedures (LSG, LAGB and RYGB). Long-term high quality studies are required to evaluate the risk of different obesity surgical procedures in children and adolescents.

In studies reporting co-morbidities variability in both the assessment and methodology is evident. There is level 2/3 evidence of improvement and resolution of co-morbidities. Paulus et al. (2015) found that over 50% of the RYGB and LGB reported resolution in associated co-morbidities, including hypertension, sleep

apnoea, insulin resistance and dyslipidaemia. Black et al. (2013) reported, 11/18 LAGB studies observed complete resolution of hypertension in 22-100% of studies, dyslipidaemia in 50% and 100% of diabetic cases after surgery. In RYGB 8/13 reported an improvement, and in LSG 4/5 studies reported resolution of comorbidities in 75-100% of studies evaluating hypertension, 56-100% of dyslipidaemia and 50-93% of those with diabetes. Inge et al. (2015) observed (level 3) an improvement in insulin sensitivity and β cell function, and metabolic improvements even with obesity persisting at one year follow-up.

Psychosocial and mental health is increasingly becoming an important parameter requiring evaluation pre and post obesity surgery. A systematic review consisting of 12 adolescent studies (Herget al., 2014) (level 2+) reported depressive symptoms ranging from 15 to 70%, anxiety symptoms 15-33% and eating disorders in 48-70%, prior to surgery. A large case series by Sysko et al. (2012) reported a significant improvement in depressive symptoms (p<0.001) at 15 months. A systematic review evaluating 10 studies (Hilstrom et al., 2015 (level 2+)) observed an overall improvement in psychosocial outcomes post operatively. Herget et al. (2014) found studies varied in evaluation in time points and no clear documentation of pharmacotherapy pre and post-surgery. Studies have observed short term improvements in psychosocial parameters, however studies have also reported a persistence of symptoms post operatively.

Zeller et al. (2011) observed an increased tendency of depressive symptoms at 18-24 months postoperatively, and Orsorio et al. (2011) observed 21.4% of patients were still suffering from clinical depressive symptoms. Overall studies (level 2, level 3) have reported improvement in quality of life parameters, physical, self-esteem domains from baseline following obesity surgery, further high level evidence is required to further evaluate the psychosocial impact upon adolescents/children pre and post-surgery.

13

Cost effectiveness

There is a lack of studies evaluating cost effectiveness of obesity surgery in children and adolescents. Aikenhead at al. (2011) in a systematic review identified three studies on LAGB in adolescents, that showed net cost saving per disability adjusted life year was \$AU4,400 (£2,092) (level 2+). Bairdain et al. (2015) (level 3) evaluated cost-effectiveness (n=11) and estimated that obesity surgery was not cost effective in the first 3 years, but cost effective after. It cost \$80,065 (£52,925) per QALY in year 4 and \$36,570 (£23,515) per QALY in year 7 (threshold of \$100,000/QALY). This small study failed to include obesity specific comorbidities, and additionally the US findings may not be entirely applicable to the UK population cohort.

What is the evidence for selection criteria and previous weight management strategies?

There is no empirical evidence of a standardised care pathway, including selection criteria. The majority of the western world follows consensus guidance, including that obesity surgery should be performed on adolescents following a multidisciplinary evaluation. Obese adolescents (\geq 40kg/m² or \geq 35kg/m² with at least one obesity associated co-morbidity) that have achieved skeletal maturation (linear growth), following failure of lifestyle interventions are considered. There is a variation and documentation in the studies in type, intensity and duration of lifestyle intervention prior to obesity surgery. The majority of obesity surgery appeared to be performed in a multidisciplinary environment. The majority of studies have excluded syndromic patients, those with severe medical or psychiatric problems and those who have disease related aetiology for obesity.

Patient participation prior to surgery provides an opportunity to evaluate behaviour and motivation. Fenning et al. (2015) pilot study (n=15) (level 3) involved two phases, firstly a 3 month preoperative program, consisting of medical examination, psychological measures, self-monitoring, physical activity, cognitive behaviour orientated therapy and psychosocial educational training, and phase two surgical phase. Phase I assessed adherence to program, parental involvement and weight loss preoperatively. They found both weight and BMI decreased over the three

months, mean loss -3.14kg/m² and the majority of patients followed the program. Interestingly, they observed poor parental participation. Compliance post operatively requires further evaluation.

Although the majority of surgery has been undertaken in non-syndromic adolescent, a recent study by Mohaidly et al. (2013) (Level 3) performed LSG on an obese 2.5 year old, and at 2 years the patient had a 27% weight loss with normalisation of BMI from 41kg/m² to 24kg/m². The authors did raise concern on parental compliance with instructions and poor attendance at follow-up. Growth, developmental and nutritional details were not included in the study. Alqahtani et al. (2015) (level -2) performed LSG on 24 patients with Prader Willi Syndrome (PWS) with a mean age of 10.7 years, observed at 5 years significant weight reduction, with rate of growth not significant between the PWS group and matched non-PWS group.

In summary, the available evidence indicates that any of these three procedures in adolescents lead to greater short-term (1-3 years) weight loss and improvements in HRQOL, psychological outcomes and comorbidities than non-invasive management alone, although there is little longer term follow-up evidence available at present. There was little evidence to indicate that one type of procedure was superior or inferior to another, and the adverse effects of obesity surgery in general are not well documented. The collection of longitudinal evidence on the short and long-term effects of obesity surgery in children, including endocrinological and metabolic effects, raises the importance of robust mechanisms to assess longer term outcomes and to ensure patients are in a position to give informed consent for the procedure.

6 Criteria for Commissioning

Surgical intervention is not generally recommended in children or young people (NICE CG189, 2014). However, obesity surgery may be considered in eligible individuals to achieve significant and sustainable weight reduction, if all the following criteria are fulfilled:

- The adolescent has been evaluated by the appropriate specialised MDT (see service specification for details) and deemed suitable appropriate for surgery.
- The adolescent has a post pubertal BMI ≥40kg/m² (BMI SD ≥3.0) or ≥35kg/m² (BMI SD ≥3.5) with significant associated comorbidities that are both predicted to have the potential to progress and are amenable to improvement/ resolution by weight loss. Obesity should have been present for several years.
- The adolescent has achieved physiological maturity (Tanner Stage 4 or above).
- The adolescent has completed clinical assessment and management treatment within a commissioned Tier 3 service.
- The decision of the MDT regarding surgery will depend on the individual's engagement and response to weight management services, their co-morbidities and risk-benefit analysis. This analysis should assess the short and long term risks of not operating versus the risks associated with surgery. In addition psychological factors, motivation/compliance, learning difficulty issues and impact on education will also be taken into account.
- The adolescent is generally fit for anaesthesia and surgery.
- The adolescent and their family commits to the need for long-term follow-up.
- Adolescents with syndromic or monogenic obesity will also be discussed by the MDT on a case by case basis and arrangements made by the MDT to seek further national expert advice/opinion on the ethical issues and supporting research.

7 Patient Pathway

Before being considered for surgery, the adolescent must have completed assessment and treatment within a commissioned Tier 3 service. It is expected that the Tier 3 service will have identified, investigated and managed the associated comorbidities prior to referral for surgical assessment to a Tier 4 service. The adequacy, intensity and duration of intervention/s will be determined by the specialist MDT: adolescents should remain in Tier 3 until all non-surgical avenues have been adequately explored and found to be unsuccessful; these approaches should be documented in the MDT discussions.

Adolescents indicated for obesity surgery should have a comprehensive clinical, psychological, educational, family and social assessment by an appropriate specialised multi-disciplinary team before undergoing surgery. This includes a full medical evaluation, and genetic screening or assessment to exclude rare, treatable causes of obesity.

Surgical care and follow-up should be coordinated around the patient and his/her family's needs, complying with the approaches outlined in the Department of Heath's 'A call to action on obesity in England' to increase attendance and compliance. Lifelong specialist follow up is advocated, with a minimum 5 year follow up care plan recommended, including transfer into the adult pathway at 18 years old where appropriate (in line with the principles outlined in the NICE guidance 'Transition from children's to adults' services' currently in development), or follow up beyond 18 years old in tertiary paediatric services or Tier 4 paediatric obesity services where adolescents are within a few years of surgery, or where local adult provision is weak.

Adolescents who have had obesity surgery should have a follow-up care plan for a minimum of 5 years within the Tier 4 obesity service, or through shared care follow up arrangements with the latter and more local specialist paediatric centres. This should include:

- monitoring nutritional intake (including protein and vitamins) and mineral deficiencies
- monitoring for comorbidities
- medication review
- dietary and nutritional assessment, advice and support
- physical activity advice and support
- psychological support tailored to the individual
- information about professionally-led or peer-support groups.

See the NHS service specification for more details.

8 Governance Arrangements

Providers, surgeons, premises, on site services and obesity surgery throughput should at least meet the IFSO Guidelines for Safety, Quality, and Excellence in Bariatric Surgery.

There must be appropriate specialised MDT composition, co-option and support, specialist professional inputs and process design for all stages of the pathway (elective and emergency). In addition, organisational arrangements for patient safety (elective and emergency) should be risk assessed, regularly tested and improved. Protocols should be audited especially the use of questionnaires for clinical assessment, generic interdisciplinary roles and substitution and/or expansion of professional roles.

The surgical service should be seamless both pre- and post-operatively with the medical specialist Tier 3/4 service, and determined by local arrangements.

The obesity surgical and medical provider will be responsible for the organisation of structured, systematic and team based follow up for a minimum of 5 years. The latter provider will make arrangements to hand over care to the adult Tier 3 service when the adolescent reaches 18 years if appropriate, with the option of continued follow up beyond 18 years of age in paediatric services or Tier 4 paediatric led services, where adolescents are within a few years of surgery, whilst transition to adult service provision is facilitated.

Follow up rates and nutritional and/or metabolic complications should be published.

9 Mechanism for Funding

Specialised complex obesity services, including obesity surgery pre-assessment, perioperative management, postoperative and longer term follow up where it occurs within the specialised service will be funded by NHS England.

10 Audit Requirements

Mandatory compliance by obesity surgery providers with National Bariatric Surgery Registry (NBSR) requirements, including 100% provision of required data, and publication of long term follow up data.

Given the relative lack of evidence relating to adverse effects (e.g. nutritional deficiencies) in the adolescent population, it would be beneficial for specific outcome requirements to be included in the NBSR dataset and published, to support longitudinal study. See specification for suggested outcome measures.

11 Documents which have informed this Policy

NHS England Clinical Commissioning Policy: Complex and Specialised Obesity Surgery NHSCB/A05/P/a.

National Institute for Health and Care Excellence Clinical Guideline 189 Obesity: identification, assessment and management of overweight and obesity in children, young people and adults..

National Institute for Health and Care Excellence Clinical Guideline 43 Transition from children's to adults' services.

12 Date of Review

This document will be reviewed when information is received which indicates that the policy requires revision.

References

Aikenhead, A.; Knai, C.; Lobstein, T.. Effectiveness and cost-effectiveness of paediatric bariatric surgery: a systematic review. Clin Obes 2011;1(1):45992.

Alqahtani, Aayed R.; Elahmedi, Mohamed O.; Al Qahtani, Awadh R.; Lee, Jaehoon; Butler, Merlin G.. Laparoscopic sleeve gastrectomy in children and adolescents with Prader-Willi syndrome: a matched-control study. Surg Obes Relat Dis 2015;0(0):0.

Bairdain, Sigrid; Samnaliev, Mihail. Cost-effectiveness of Adolescent Bariatric Surgery. Cureus 2015;7(2):e248.

Black, JA; White, B; Viner, RM; Simmons RK. Bariatric surgery for obese children and adolescents: a systematic review and meta-analysis. Obes Rev. 2013 Aug;14(8):634-44.

Fennig, Silvana; Brunstein-Klomek, Anat; Sasson, Ariel; Halifa Kurtzman, Irit; Hadas, Arie. Feasibility of a Dual Evaluation/Intervention Program for Morbidly Obese Adolescents. Isr J Psychiatry Relat Sci 2015;52(2):107-112.

Herget, Sabine; Rudolph, Almut; Hilbert, Anja; Blüher, Susann. Psychosocial status and mental health in adolescents before and after bariatric surgery: a systematic literature review. Obes Facts 2014;7(4):233-245.

Hillstrom, Kathryn A.; Graves, Joyce K.. A review of depression and quality of life outcomes in adolescents post bariatric surgery. J Child Adolesc Psychiatr Nurs 2015;28(1):50-59.

Inge, Thomas H.; Zeller, Meg H.; Jenkins, Todd M.; Helmrath, Michael; Brandt, Mary L.; Michalsky, Marc P.; Harmon, Carroll M.; Courcoulas, Anita; Horlick, Mary; Xanthakos, Stavra A.; Dolan, Larry; Mitsnefes, Mark; Barnett, Sean J.; Buncher, Ralph; Teen-LABS Consortium. Perioperative outcomes of adolescents undergoing bariatric surgery: the Teen-Longitudinal Assessment of Bariatric Surgery (Teen-LABS) study. JAMA Pediatr 2014;168(1):47-53.

Inge, Thomas H.; Prigeon, Ronald L.; Elder, Deborah A.; Jenkins, Todd M.; Cohen, Robert M.; Xanthakos, Stavra A.; Benoit, Stephen C.; Dolan, Lawrence M.; Daniels, Stephen R.; D'Alessio, David A.. Insulin Sensitivity and β-Cell Function Improve after Gastric Bypass in Severely Obese Adolescents. J. Pediatr. 2015;0(0):0.

Jones Nielsen, JD; Laverty, AA; Millett, C; Mainous, AG; Majeed, A; Saxena, S. Rising obesity-related hospital admissions among children and young people in England: national time trends study.

PLoS One. 2013 Jun 12;8(6):e65764. doi: 10.1371/journal.pone.0065764. Print 2013.

Lennerz, B. S.; Wabitsch, M.; Lippert, H.; Wolff, S.; Knoll, C.; Weiner, R.; Manger, T.; Kiess, W.; Stroh, C.. Bariatric surgery in adolescents and young adults--safety and effectiveness in a cohort of 345 patients. Int J Obes (Lond) 2014;38(3):334-340.

Mohaidly, Mohammed AI; Suliman, Ahmed; Malawi, Horia. Laparoscopic sleeve gastrectomy for a two-and half year old morbidly obese child. Int J Surg Case Rep 2013;4(11):1057-1060.

O'Brien, PE; Sawyer, SM; Laurie, C; Brown, WA; Skinner, S; Veit, F; Paul, E; Burton, PR; McGrice, M; Anderson, M; Dixon, JB. Laparoscopic adjustable gastric banding in severely obese adolescents: a randomized trial. JAMA. 2010 Feb 10;303(6):519-26.

Osorio, A; Moreira-Pinto, J; Pereira, J; Silva, G; Bonet, B; Cidade-Rodrigues, JA; Ferreira-de-Sousa, JA; Enes, C; Mansilha, H. 9 Years after the first laparoscopic adjusted gastric banding (LAGB) in adolescents: the Portuguese experience. Eur J Pediatr Surg. 2011 Oct;21(5):331-4. doi: 10.1055/s-0031-1277211. Epub 2011 Aug 12.

Oude Luttikhuis, Hiltje; Baur, Louise; Jansen, Hanneke; Shrewsbury, Vanessa A.; O'Malley, Claire; Stolk, Ronald P.; Summerbell, Carolyn D.. Interventions for treating obesity in children. Cochrane Database Syst Rev 2009;0(1):CD001872.

Paulus, Givan F.; de Vaan, Loes E. G.; Verdam, Froukje J.; Bouvy, Nicole D.; Ambergen, Ton A. W.; van Heurn, L. W. Ernest. Bariatric surgery in morbidly obese adolescents: a systematic review and meta-analysis. Obes Surg 2015;25(5):860-878.

Pedroso, Felipe E.; Gander, Jeffery; Oh, Pilyung Stephen; Zitsman, Jeffrey L.. Laparoscopic vertical sleeve gastrectomy significantly improves short term weight loss as compared to laparoscopic adjustable gastric band placement in morbidly obese adolescent patients. J. Pediatr. Surg. 2015;50(1):115-122.

Sysko, R; Devlin, MJ; Hildebrandt, TB; Brewer, SK; Zitsman, JL; Walsh, BT. Psychological outcomes and predictors of initial weight loss outcomes among severely obese adolescents receiving laparoscopic adjustable gastric banding. J Clin Psychiatry. 2012 Oct;73(10):1351-7. doi: 10.4088/JCP.12m07690.

Zeller, MH;, Reiter-Purtill, J; Ratcliff, MB; Inge, TH; Noll, JG. Two-year trends in psychosocial functioning after adolescent Roux-en-Y gastric bypass. Surg Obes Relat Dis. 2011 Nov-Dec;7(6):727-32. doi: 10.1016/j.soard.2011.01.034. Epub 2011 Mar 5.